\documentclass[letterpaper,12pt]{article}
\usepackage[utf8x]{inputenc}
\usepackage{amsmath}
\usepackage[letterpaper,margin=1in,includehead=true]{geometry}
\usepackage{comment}
\usepackage{graphicx}
\usepackage{fancyhdr}
\usepackage{enumerate}
\pagestyle{fancy}
\usepackage{color}
\usepackage{wrapfig}
%\usepackage{enumitem}
\usepackage{tikz, tikz-3dplot, pgfplots}
\usetikzlibrary{calc}
%\usepackage{titling}
%\setlength{\droptitle}{-1.5in}

%opening
\title{\vspace*{-1.5in}\textbf{Math 2300-007: \S6.1 - More About Areas}\\[1ex]\normalsize(With thanks to Faan Tone Liu)}
\date{}
%\author{}

\begin{document}

\maketitle
%\underline{Math 1300-400: Quiz 1}
\lhead{Math 2300-007: Calculus II}
\rhead{February 5, 2018}
\chead{\S5.7 - More About Areas}
\vspace{-.6in}
\noindent
Complete the following problems about areas. As you work, think about the key points, so can summarize them in question 4. 
\vskip12pt
\begin{enumerate}
\item Find the area bounded between $y= 5x-x^2$ and $y=x$.\vfill
\item Find the area of the region bounded by $y=4x-x^2$, $y=4-x$, and the $x$-axis.\vfill
\newpage
\item Set up the integral to find the area bounded between $x=2y^2$ and $x=27-y^2$.
\vfill
\item What advice would you give to another student for finding areas between curves? What are some of the steps? The key points? \vfill
\end{enumerate}
\end{document}
